Author's response to reviews

Title: A 65-year-old woman presenting with lung mass and pericardial effusion: a case report

Authors:

George A Margaritopoulos (gmargaritop@yahoo.gr)
Athanasia Proklou (proklath@yahoo.gr)
Eleni Lagoudaki (elagoudakimd@gmail.com)
Argiro Voloudaki (voloudak@med.uoc.gr)
Nikos M Siafakas (siafak@med.uoc.gr)
Katerina M Antoniou (katerinaantonio@yahoo.gr)

Version: 3 Date: 1 March 2012

Author's response to reviews:

Comment: Please include a figure title and legend section after the reference list

Reply: We have added a figure legend section after the references and a figure title for each one of the four figures which appear in the manuscript. We have also erased the legends from the figures which are submitted as separate files.

best regards

Dr George Margaritopoulos